Author's response to reviews

Title: The time since last menstrual period is important as a clinical predictor for non-steroidal aromatase inhibitor-related arthralgia

Authors:

Mitsunori Sasa (breast@mb.tcn.ne.jp)
Miyuki Kanematsu (miyuki@higasitokusima.hosp.go.jp)
Masami Morimoto (hmm-morimoto@ezweb.ne.jp)
Junko Honda (jhonda@higasitokusima.hosp.go.jp)
Taeko Nagao (tnagao@clin.med.tokushima-u.ac.jp)
Misako Nakagawa (kiramisa812@clin.med.tokushima-u.ac.jp)
Masako Takahashi (takapy55@tune.ocn.ne.jp)
Akira Tangoku (tangoku@clin.med.tokushima-u.ac.jp)

Version: 4 Date: 30 September 2011

Author's response to reviews:

The BMC Series Editorial Production Team
2011/09/30

Dear Sir:

Thank you for your kind mail of 2011/9/30, and comments to our 2nd-revised paper entitled “MS: 7448794175801567 The Time Since Last Menstrual Period Is Important as a Clinical Predictor for Non-steroidal Aromatase Inhibitor-Related Arthralgia” from the editor. We appreciate the additional comments and are sorry for causing you trouble. We wish our 3rd-revised paper would be published after minor revision.

We are submitting our manuscript, 3rd-revised in accordance with the various, helpful suggestions of the editor.

According to your suggestions, we rewrote these several points as follows.

Title was changed to “The time since last menstrual period is important as a clinical predictor for non-steroidal aromatase inhibitor-related arthralgia”.

The name of the authors were confirmed
Email address for authors was corrected.
Tables 1-3 were included to the manuscript following the references.
Figure was corrected according to the figure guide.

We hope all these corrections are acceptable and rapid publication of our article.

Your kind consideration of our paper would be greatly appreciated.

Yours faithfully,
Mitsunori Sasa, MD
<p.s.>
All correspondence concerning this paper
should be addressed to:
Mitsunori Sasa, MD
Department of Surgery, Tokushima Beast Care Clinic,
4-4-7, Nakashimada-cho, Tokushima, Japan, 770-0052
mail: breast@mb.tcn.ne.jp fax:088-633-8485